Author’s response to reviews

Title: The atrial uremic cardiomyopathy regression in patients after kidney transplantation. The prospective echocardiographic study

Authors:

Tomasz Zapolski (zapolia@wp.pl)
Jacek Furmaga (jacekf61@o2.pl)
Andrzej Wysokiński (a.wysokinski@umlub.pl)
Anna Wysocka (dojaty@gmail.com)
Sławomir Rudzki (slawomir.rudzki@umlub.pl)
Andrzej Jaroszyński (jaroszynskiaj@interia.pl)

Version: 2  Date: 17 Mar 2019

Author’s response to reviews:

Dear Professor Mitra Mahdavi-Mazdeh

Thank you for kind consideration for publication our manuscript (BNEP-D-19-00002) in the BMC Nephrology. Accordingly to suggestions enclosed in Reviewers commentary we revised the manuscript and we made some changes as follows:

Adarsh Babu, MBBS,MRCP. (Reviewer 2):

1. Minor language corrections

ad. 1. The entire text has been re-checked by a professional English language editing service dealing with medical translations.

The language and spelling correction, introduced by a professional English language editing service, were marked in a red font.

We hope that our changes have improved quality of presented manuscript and thereby you will be able to consider this version of paper for publication in the BMC Nephrology.
With compliments –

Tomasz Zapolski, Jacek Furmaga, Andrzej Jaroszyński, Anna Wysocka, Sławomir Rudzki, and Andrzej Wysokiński